

Reporting radiographers' interaction with Artificial Intelligence—How do different forms of AI feedback impact trust and decision switching?

Rainey, C., Bond, R., McConnell, J., Hughes, C., Kumar, D., McFadden, S., & Lourentzou, I. (Ed.) (2024). Reporting radiographers' interaction with Artificial Intelligence—How do different forms of AI feedback impact trust and decision switching? *PLOS Digital Health*, *3*(8), 1-22. Advance online publication. https://doi.org/10.1371/journal.pdig.0000560

Link to publication record in Ulster University Research Portal

Publication Status: Published online: 07/08/2024

DOI: 10.1371/journal.pdig.0000560

Document Version Publisher's PDF, also known as Version of record

Document Licence: CC BY

General rights

The copyright and moral rights to the output are retained by the output author(s), unless otherwise stated by the document licence.

Unless otherwise stated, users are permitted to download a copy of the output for personal study or non-commercial research and are permitted to freely distribute the URL of the output. They are not permitted to alter, reproduce, distribute or make any commercial use of the output without obtaining the permission of the author(s)

If the document is licenced under Creative Commons, the rights of users of the documents can be found at https://creativecommons.org/share-your-work/cclicenses/.

Take down policy The Research Portal is Ulster University's institutional repository that provides access to Ulster's research outputs. Every effort has been made to ensure that content in the Research Portal does not infringe any person's rights, or applicable UK laws. If you discover content in the Research Portal that you believe breaches copyright or violates any law, please contact pure-support@ulster.ac.uk



Citation: Rainey C, Bond R, McConnell J, Hughes C, Kumar D, McFadden S (2024) Reporting radiographers' interaction with Artificial Intelligence —How do different forms of Al feedback impact trust and decision switching? PLOS Digit Health 3(8): e0000560. https://doi.org/10.1371/journal.pdig.0000560

Editor: Ismini Lourentzou, University of Illinois Urbana-Champaign, UNITED STATES OF AMERICA

Received: February 22, 2024

Accepted: June 22, 2024

Published: August 7, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pdig.0000560

Copyright: © 2024 Rainey et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because of ethical restrictions on sharing sensitive data. Access to the data can be

RESEARCH ARTICLE

Reporting radiographers' interaction with Artificial Intelligence—How do different forms of AI feedback impact trust and decision switching?

Clare Rainey^{1*}, Raymond Bond², Jonathan McConnell³, Ciara Hughes¹, Devinder Kumar^{4*}, Sonyia McFadden¹

Ulster University, School of Health Sciences, York St, Belfast, Northern Ireland, 2 Ulster University, School of Computing, York St, Belfast, Northern Ireland, 3 NHS Leeds Teaching Hospitals, Leeds, United Kingdom,
School of Medicine, Stanford University, California, United States of America

¤ Current address: Head-MLOps, Layer6 AI

* c.rainey@ulster.ac.uk

Abstract

Artificial Intelligence (AI) has been increasingly integrated into healthcare settings, including the radiology department to aid radiographic image interpretation, including reporting by radiographers. Trust has been cited as a barrier to effective clinical implementation of AI. Appropriating trust will be important in the future with AI to ensure the ethical use of these systems for the benefit of the patient, clinician and health services. Means of explainable AI, such as heatmaps have been proposed to increase AI transparency and trust by elucidating which parts of image the AI 'focussed on' when making its decision. The aim of this novel study was to quantify the impact of different forms of AI feedback on the expert clinicians' trust. Whilst this study was conducted in the UK, it has potential international application and impact for AI interface design, either globally or in countries with similar cultural and/or economic status to the UK. A convolutional neural network was built for this study; trained, validated and tested on a publicly available dataset of *MU*sculoskeletal *RA*diographs (MURA), with binary diagnoses and Gradient Class Activation Maps (GradCAM) as outputs. Reporting radiographers (n = 12) were recruited to this study from all four regions of the UK. Qualtrics was used to present each participant with a total of 18 complete examinations from the MURA test dataset (each examination contained more than one radiographic image). Participants were presented with the images first, images with heatmaps next and finally an AI binary diagnosis in a sequential order. Perception of trust in the AI systems was obtained following the presentation of each heatmap and binary feedback. The participants were asked to indicate whether they would change their mind (or decision switch) in response to the AI feedback. Participants disagreed with the AI heatmaps for the abnormal examinations 45.8% of the time and agreed with binary feedback on 86.7% of examinations (26/30 presentations).'Only two participants indicated that they would decision switch in response to all AI feedback (GradCAM and binary) (0.7%, n = 2) across all datasets. 22.2% (n = 32) of participants agreed with the localisation of pathology on the heatmap. The level of agreement with the GradCAM and binary diagnosis was found to be correlated with trust

provided following a successful application to Ulster University's Nursing and Health Research Ethics Filter Committee. Ulster University's Research Portal contains metadata on the dataset and instructions on how to request access to this dataset. This information can be accessed at https://doi.org/10.21251/50890091-4b54-4644b980-ea9da646aa0e.

Funding: This work has been funded by the College of Radiographers Research Industry Partnership Research awards scheme (CoRIPS) no. 183. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

(GradCAM:—.515;—.584, significant large negative correlation at 0.01 level (p = < .01 and -.309;—.369, significant medium negative correlation at .01 level (p = < .01) for GradCAM and binary diagnosis respectively). This study shows that the extent of agreement with both AI binary diagnosis and heatmap is correlated with trust in AI for the participants in this study, where greater agreement with the form of AI feedback is associated with greater trust in AI, in particular in the heatmap form of AI feedback. Forms of explainable AI should be developed with cognisance of the need for precision and accuracy in localisation to promote appropriate trust in clinical end users.

Author summary

Artificial Intelligence (AI) is pervading many areas of our day-to-day lives, including healthcare. In radiology, AI models are being used to assist in the interpretation of radio-logical images. Professional communities are aware of the impressive performance of these systems but less aware of the relationship between AI and the human end-user. The expert/experienced users' perception of trust in these systems has been cited as a major barrier to the successful integration of healthcare AI. This study uses 'reporting radiographers', who provide written diagnoses on plain radiographic images, to investigate the impact of the type of AI feedback offered on the expert users' trust perception. Participants were presented with two commonly used forms of AI feedback–a binary diagnosis, with AI confidence indication, and a visual form of feedback ('heatmap') to indicate the area of the image the AI found most important in reaching its decision. The study found that agreement with the AI feedback is an important indicator of trust, particularly with the 'heatmap'. This is important in the design of the user-interface in clinical AI, where acceptance of the form of feedback may prove to be a barrier or a facilitator to the integration of AI into clinical practice.

1 Introduction

Studies investigating the potential of 'Artificial Intelligence' (AI) as a human adjunct in detecting pathology on radiographic images began in the 1960s with a 'proof of concept' system converting images to numerical data [1]. Since then, electronic evolutionary changes with rapidly increasing computational power availability has permitted the development of ever more sophisticated detection systems. Differing methods of analysis of medical, and other, images have been developed, with Deep Learning (DL) using Convolutional Neural Networks (CNN) being the most recent and seemingly most promising form of AI for detecting disease from radiographic images. CNNs are trained to perform certain specific tasks and develop accuracy based on feedback. The algorithm therefore develops its performance ability rather like a human brain with layers of 'neural networks' built up to allow improvement with experience. This is referred to as supervised Machine Learning (ML) where a large set of pre-labelled cases are used to 'teach' the computer to discriminate between normal and abnormal cases.

These developments have been the focus of much media and professional attention. The development of AI has been targeted as an area of focus for modernising and future-proofing the National Health Service in the UK [2]. There are several significant drivers to the development of AI as a tool in the health care setting; namely, time constraints / efficiency, error avoidance or minimisation and workflow augmentation [3,4]. It is estimated that the implementation of an effective AI system for automated image reporting could reduce the time that

radiologists spend reviewing images by 20% [5], and thus liberate 890,000 hours of radiologist time per annum [5]. This time can be spent doing non automatable tasks such as providing personalised patient care and more complex tasks where human input is essential. This is particularly important in worldwide healthcare systems, coping with current multi-faceted pressures, including the exacerbation of these resulting from the COVID-19 pandemic, where resources are limited [6].

Despite reported accuracies of AI for a number of applications in radiology [7,8,9], clinicians' trust in AI remains a barrier to implementation in the health care setting [10]. This is particularly the case with the use of DL. As mentioned, DL algorithms make use of multiple neural layers to analyse and process image data but there are a number of these layers which are hidden to the user. It is not apparent, therefore, how the algorithm reaches its ultimate decision. This, not only, has implications for users' trust in these systems but should be central when considering the ethical issues surrounding the clinical use of AI in radiology [11]. Attempts are currently being made to make this process more transparent by the use of visual representations of the areas of attention of the AI, for example, heat or attention maps [12,13,14].

Plain radiographs of the appendicular skeleton are commonly performed examinations, with 23.2 million requested in NHS England per annum [15], therefore clinicians are accustomed to using them in the clinical setting. Despite plain radiography being commonly used in clinical practice, the use of AI in identification of abnormality on appendicular skeletal radiographs remains a relatively unexplored area, with the first publication of experimental results in 2017 [16]. Other studies followed, reporting impressive accuracies [17].

This proposed study aims to use a larger dataset of projections for seven anatomical areas, that includes fingers, hand, wrist, elbow, forearm, shoulder and humerus, in order to better replicate the range of patient presentation in the clinical setting.

Radiographer reporting is an established practice in the UK, however the NHS Diagnostics: Recovery and Renewal report (2020) [18] calls for this role to be further expanded to allow 50% of all plain radiographic images to be reported on by a radiographer. Whilst this is a welcome recommendation, the current dearth of radiographers and radiologists in the UK may mean that, not only must the number of radiology professionals (radiographers and radiologists) being trained increase, but efficiencies must be found, for example, the integration of AI to assist with faster reporting turnaround times [18,19,20,21] as suggested by NHS England draft document of February 2023 titled 'Diagnostic Imaging Reporting Turnaround Times [22]

This study attempts to quantify expert radiographers' trust in AI during and after exposure to AI binary diagnoses and heat maps on plain radiographs of the appendicular skeleton.

2 Aim and objectives

This study's principal aim was to investigate the factors impacting reporting radiographers trust in Artificial Intelligence (AI) for use in image interpretation. Musculoskeletal radiographs of the upper extremity from the MURA dataset [23] were used.

The objectives of this study were to:

- i. investigate reporting radiographers' trust in an AI system used to provide clinical decision support on radiographic images of the upper appendicular skeleton.
- ii. investigate the factors impacting trust in these systems, including agreement with heatmaps, used as means of explainable AI and binary diagnosis from the AI.
- iii. determine the propensity of the users to decision switch following provision of the AI feedback

3 Materials and methods

3.1 Ethical permission

Ethical permission was granted from Ulster University Nursing and Health Research Filter Committee (FCNUR-20-026).

3.2 Model training

The MURA dataset is a publicly available dataset used for training AI models for computer vision tasks in pathology detection on plain radiographs of the upper appendicular skeleton. This ran as a 'competition' where competitors submit their code to a central repository (https://codalab.org/) for testing on an unseen test set. The dataset consists of 40,561 radio-graphic images (14,863 examinations, training dataset n = 36,808 images) which have ground truth established (normal/abnormal). The dataset is split into training and validation datasets. The training set was used for initial training of the model and the validation set was used to fine-tune the parameters until an acceptable accuracy was reached.

The images in the dataset vary in quality, resolution, aspect ratio. The images are exclusive to each dataset, with no overlap.

3.3 Test dataset

The test set of the MURA competition is not publicly available. For this study half the validation set was used as a test set, consisting of 783 patients, 1,199 studies and 3197 images. The remainder was used as a 'validation set'. The MURA competition has now closed and, therefore, there were no established diagnoses available for the testing and validation sets, however, as the participants were practicing reporting radiographers, their consensus diagnosis was taken as 'ground truth'.

No overlap between any of the sets was permitted. The test set was chosen to contain approximately half of each of the seven upper extremities for adequate and balanced representation of each class.

3.4 AI model

A pretrained (ImageNet) ResNet-152 convolutional neural network (CNN) was used in this study. For each examination, the arithmetic mean of the output is used to determine pathology, with any probability over 0.5 deemed abnormal. The model was trained on the training set until no further improvement was observed. Adam optimiser was used with initial learning rate to 10^-4.

3.5 Salience map

A saliency map was produced for each image to 'explain' the AI prediction. This was created using a technique used by one of the authors in a previous study [12]. The whitest area of the colour map indicated the strongest areas of spatial location used by the model to determine the probability. Null values are represented as black.

3.6 Patient-public involvement

A patient-public involvement group was established to advise on the relevance of this study for key stakeholders. Feedback was also sought on the methodology of the study and plans for dissemination, including the appropriateness of presentation at conferences and publishing the work in international, peer reviewed journals [24]. The group consisted of patients and

radiographers (n = 5). The background, rationale and expected impact of this study was discussed over one face to face meeting and subsequent email updates.

3.7 Pilot study

The study was piloted amongst three experienced radiographers, two from a clinical reporting background and one from an academic institution, with extensive experience in plain radiography. Each of the pilot participants had in excess of 20 years clinical and/or academic experience.

A random sample of radiographs not included in the main study were selected and embedded into the Qualtrics platform. Participants were asked to provide feedback on the suitability of the questions asked, the accessibility of survey platform, time taken for completion and the quality of the radiographic images, thus ensuring both face and content validity [25]. As a result of this, the number of images which would be included in the main study was decided and based on an approximation of the number of images that could be interpreted in a maximum of 90 minutes. Participants in the pilot study reported that they found the study easy to follow and comprehend. No changes were made to the study flow in Qualtrics as a result of the pilot study, however, image quality was noted as an important issue. Participants reported that many of the images in the dataset were of low diagnostic quality to the human eye. A question relating to the participants' perception of the image quality of each presented image was then included in the final version of the study to ensure all examinations included in the datasets were acceptable for interpretation by the participants.

3.8 Recruitment / Participant selection

A request for reporting radiographers to participate in this study was made through authors' professional networks including reporting and advanced practice special interest groups (The Society of Radiographers (SoR) Consultant Radiographers Advisory Group and The SoR Reporting Radiographers Special Interest group). Interested participants were directed how to contact the researcher (CR) by email. The first three interested participants in each region of the UK (England, Scotland, Wales and Northern Ireland) were recruited to the study. A £25 gift voucher was offered following expression of interest to take part in the study to encourage completion of the study, therefore minimising the risk of attrition. All participants completed the entire study to completion.

Each participant was emailed a unique link to the survey on the 18th July 2022 and a reminder email sent on the 2nd August 2022, resulting in full participation. A unique link was provided to ensure individuality of responses to confirm the resultant data met the requirements to allow for analysis of interrater agreement using Fleiss kappa [26].

A power calculation was not carried out as this is a novel study with lack of comparable quantitative research in the area. The literature indicates that in these cases a sample size of 10–15 participants is adequate for this type of study [27,28].

3.9 Construction of the dataset for Qualtrics

Each imaging series in each case of the MURA test set were initially analysed by one of the authors not involved in the interpretation (CR) and were excluded from this study if the image contained surgical hardware, for example, following internal surgical fixation of trauma or any artefact felt to be significantly obscuring anatomy as demonstrated in Fig 1. The overall diagnostic quality of the images included in the main study were then reviewed by the remining authors with experience in radiographic image interpretation (SMF and JMcC) and rejected as necessary (Figs 2 and 3). The rationale of each exclusion was tabulated and is available, upon request, from the corresponding author (CR). Following exclusions, 299 imaging series were



Fig 1. Example of rejected image due to presence of internal fixation. https://doi.org/10.1371/journal.pdig.0000560.g001



Fig 2. Example of rejected image due to poor image quality. https://doi.org/10.1371/journal.pdig.0000560.g002





included in this study. Each imaging series contains one or more projections of the anatomical area. Each imaging series was allocated a number from 1–299. A random number generator (Random.org) was used to select 110 examinations. Each examination was then closely inspected by the author (CR) for diagnostic acceptability and verified by another two authors involved in the pilot study (SMF, JMcC).

The remaining 30 full examinations (minimum two images in each examination) were divided into two datasets of 12, and one dataset of six examinations. Each participant was randomly allocated one of the datasets of 12 examinations and all participants were presented with the dataset of six examinations, to allow the author to compute the interrater reliability across all participants (S1 File).

Each image was adjusted to optimise viewing by modifying light, clarity and colour in Microsoft Photos and saved as a.png image. This is a lossless compression format and has been used in a similar study [14]. Three authors independently reviewed each image for quality assurance purposes (CR, SMF, JMcC).

The radiographic images were randomly assigned to each dataset and embedded in the Qualtrics survey platform. Each dataset (DS) was assigned a letter (a, b or c) and each examination a number (1–12 for DS a and b or 1–6 for DS c), for example, DS a: examination 1, DS a, examination 2, and so on. Each participant was randomly allocated one of dataset (DS) a or b using the randomiser function in Qualtrics. Each participant was allocated DS c. In total, each participant was asked to interpret 18 radiographic examinations.

Each of the participants taking part in the study received their own individual link to the study. Internet Protocol (I.P.) addresses were not gathered to ensure full anonymity. Participants were provided with an information and instruction sheet via email. Included in this were contact details of the Research Governance team at the institution. The purpose for the data collection, means of analysis and dissemination strategy was provided. Consent was indicated and recorded by the participants on an initial slide in the Qualtrics platform used for this study. If participants indicated that they did not consent to taking part, they were exited from the survey and no data was collected. No participant indicated that they did not consent.

Participants were asked to indicate their binary diagnosis and localisation of any pathology on each image, before and after provision of a Gradient-weighted Class Activation Map (GradCAM) 'heatmap' overlay. They were then asked to indicate the extent of the heatmap agreement with their localisation of pathology (if applicable). They were asked to indicate their trust in the system following provision of the heatmap. All the images in the examination were presented again to the participant, along with the binary diagnosis (pathology/no pathology) from the AI. Participants were asked to indicate their level of trust in the AI system following both heatmap and binary feedback. They were then asked if the AI feedback would have caused them to change their mind from their initial decision. Finally, they were asked to rate the diagnostic quality of the images included on a scale of zero to five (S1 File).

3.10 Data analysis

Interrater agreement across all readers on all datasets was established using Fleiss' kappa to determine the level of agreement across two or more raters [26].

Descriptive statistics describe the agreement of the AI and the user, using consensus diagnosis of the participants as ground truth. Consensus was determined as the most popular decision of the participants (S2 File). Percentage agreement with the AI diagnosis was established.

Ordinal and nominal perception data was obtained to gain an insight into the users' trust, therefore non-parametric tests were used for correlation. Spearman's rho (r_s) and Kendall's kau (τ_b) were used to investigate any correlation between:

- 1. Trust (0–5) and level of agreement with the heatmap feedback from the AI (perfect agreement (coded 1), partial agreement (2) and disagreement (3))
- Trust (0-5) and level of agreement with the binary feedback from the AI (perfect agreement (1), partial agreement (2) and disagreement (3)
- 3. Trust (0–5) and propensity of the user to change their mind from their initial decision (yes (1), maybe (2), no (3)
- 4. Image quality (0–5) and agreement with binary AI feedback (perfect agreement (1), partial agreement (2) and disagreement (3)

5. Image quality (0-5) and trust (0-5)

Image quality (0 representing no trust– 5 representing absolute trust) and propensity of the user to change their mind from their initial decision (Yes (1), maybe (2), no (3))

6. Level of agreement with AI was possible only on images where abnormality was detected. Data entries indicating that there was no abnormality, as determined by the consensus diagnosis of the participants, were coded as 'missing data' before analysis.

The strength of any relationship was determined using the suggestion by Cohen (1988), cited in Pallant, 2007 [29]:

Small .01 to .29 Medium .30 to .49 Large .50 to 1.0

4 Results

4.1 Participants

There were 12 participants in this study, three from each region of the UK, namely, England, Scotland, Wales and Northern Ireland. Each participant was a practicing reporting radiographer, a registered radiographer who has received extra postgraduate academic and clinical training to provide written diagnoses on radiographic images.

4.2 Interrater agreement

Interrater agreement was established on all examinations within each of the datasets. Fleiss' kappa (k) was calculated and indicated:

moderate agreement between participants allocated dataset (DS) a (n = 5, k = .581, 95% C.I. .475 - .688, p < .05),

moderate agreement between participants allocated DS b (n = 7, k = .410, 95% C.I. .338 - .482, p < .05), and fair agreement between participants allocated DS c (n = 12, k = .228, 95% C. I. .172 - .283, p < .05). Agreement was calculated for individual categories (abnormal and normal). The magnitude of agreement was determined by levels determined for the Cohen's kappa coefficient, as suggested by Altman (1999 [30]):

Poor < .02 Fair .21 to .40 Moderate .41 to .60 Good .61 to .80 Very good .81 to 1.00 No statistically signifi

No statistically significant difference in responses was found for any dataset, concluding that participants responded in a similar way, therefore descriptive statistics below are reported per image and per examination, rather than per participant.

4.3 Agreement of participants with AI heatmap

Following presentation of the heatmap for each of the images in each examination the participant was asked to indicate their level of agreement with the heatmap localisation. This was only relevant when the participant deemed there was abnormality present on the image. Level of agreement was measured by asking the participant to choose from one of the following options:

· Yes, in all areas I have previously identified

- Yes, in more than half of the areas I have previously identified
- · Yes, in less than half of the areas I have previously identified
- No, there are no areas of agreement with areas I have previously identified
- · No pathology

These findings were coded into four categories for analysis:

- Perfect agreement (Yes, in all areas I have previously identified, coded '1')
- Partial agreement (yes, in more than half and less than half, coded '2')
- Disagreement (no, there are no areas of agreement, coded '3') and
- No pathology.

The responses indicating there was no pathology were coded as missing data and excluded from the analysis.

Following removal of the 'no pathology' responses, there were 144 decision points remaining. There was perfect agreement with the heatmap noted in 22.2% of images (n = 32), partial agreement in 31.9% of images (n = 46) and disagreement in 45.8% of images (n = 66). Data is presented in full in Figs 4 and 5.

4.4 Agreement of participants with binary AI diagnosis

Initially, agreement of the participants with the AI was established by consensus diagnosis for each examination (S2 File). The participants agreed with the binary AI feedback on 86.7% of the examinations (n = 26 out of 30 examinations). Of those where the consensus diagnosis of the participants differed from the AI binary feedback, three were false positive, where the AI determined pathology in examinations where the radiographer did not and one false negative, where the AI failed to detect a pathology identified by the participant (Fig 6). Image quality was rated below the mean image quality (3.1) of the study for all these examinations (2.6, 3.0, 2.1 and 2.0 for Patients 3a, 5a, 3b and 5b respectively).

Further analysis per participant for each examination indicated that there was agreement with the AI, either perfect or partially, for all examinations except for examination 3b, where all participants disagreed with the AI decision (100%, n = 7 participants) and examination 5b, where there was disagreement with the AI indicated by most of the participants (86%, n = 6 participants) (Fig 6)).

Overall, there was agreement with the AI binary diagnoses in 66.9% of instances (i.e., n = 147 out of a total of 216 decision points (12 participants' decisions over 30 examinations)), partial agreement 22.5% (n = 46) and no agreement in 11.2% of examinations (n = 23). Full detail is presented in Figs 7 and 8, including indication of the instances of disagreement with the AI decision.

4.5 Decision switching

Decision switching is the self-reported likelihood of the participants to change their mind from their initial diagnosis. Detailed findings, per dataset, are presented in Figs 9 and 10. There were 60 decision points in response to this question for each dataset (n = 5 participants for DSa (12 examinations), 7 for DSb (12 examinations) and 12 for DSc (6 examinations)), resulting in a total of 216 decision points overall. Of the total 216 responses to this question across all datasets there were only two instances (examination 11b and 1c) where participants indicated that they would change their mind in response to the AI feedback (1.1%), n = 35



Fig 4. Participant agreement with heatmap feedback.

(15.7%) indicated that they might change their mind from their initial diagnosis and the remainder (n = 179, 83.5%) indicated that they would retain their initial decision.

4.6 Image quality

In response to the feedback to the pilot study regarding poor diagnostic quality of some of the images, participants were asked to rate the quality of the images in each examination with 0 indicating an undiagnostic image and 5, perfect image quality. The mean score across all images in this study was 3.1.

4.7 Correlation analysis (Table 1)

Users' trust perception in AI was obtained after the heatmap presentation for each radiographic image and after the presentation of all images (including heatmaps and binary feedback) in each examination (S1 File).



Fig 5. Participant agreement with heatmap feedback (pathological images only).

There was a statistically significant large negative correlation between users' trust, per image (0–5) and the level of agreement with the heatmap for each image (perfect, partial and no agreement) (n = 645, $\tau_b = -.515$, $r_s = -.584$, p = < .01)

There was also a statistically significant moderate negative correlation between trust, obtained at the end of the examination (following all images, heatmaps and binary AI feedback) and level of agreement with the binary feedback from the AI (perfect agreement, partial agreement and disagreement) (n = 216, τ_b = -..309, r_s = -..369, p = < .01)

A statistically significant small negative correlation was also found between the users' perception of image quality and agreement with binary AI feedback at the end of the examination (n = 126, τ_b = -.238, r_s = -.268, p = < .01). Finally, a statistically significant small positive correlation was found between image quality and trust (n = 216, τ_b = .219, r_s = .256, p = < .01)

Patient 3a All images from this examination are given again below. The Al system determined that this examination/imaging series DID NOT contain evidence of pathology. Do your initial diagnoses agree with this?

Patient 5a All images from this examination are given again below. The Al system determined that this examination/imaging series DID contain evidence of pathology. Do your initial diagnoses agree with this?



R 7MS R

Patient 5b: All images from this examination are given again below. The Al system determined that this examination/imaging series DID contain evidence of pathology. Do your initial diagnoses agree with this?

Patient 3b: All images from this examination are given again below. The Al system determined that this examination/imaging series DID contain evidence of pathology. Do your initial diagnoses agree with this?





Fig 6. Instances of human/AI diagnostic disagreement with AI Binary feedback.

https://doi.org/10.1371/journal.pdig.0000560.g006

There were no statistically significant correlations between trust and propensity to change their mind from the initial decision, or between the quality of images and propensity to change their mind (Full detail is provided in Table 1).

In summary, as agreement with the heatmap and binary AI feedback increases, trust increases. With this group of participants, the decision switching rate does not seem to be correlated with trust in the AI. As the quality of image increases, the users' agreement with AI feedback and trust perception increases. There is no correlation found between image quality and decision switching rate with this participant cohort.

5 Discussion

5.1 Participants and interrater agreement

There was fair to moderate agreement across all participants in this study, in all included examinations (n = 30). All participants were allocated DS c and therefore this may have contributed to the increased variability in responses.

5.2 Agreement of participants with AI heatmap

Methods of explainability have been suggested as a way to calibrate trust and ensure responsible use of AI systems, i.e., allowing the user to interrogate the system to make a decision on what their trust level should be on a case-by-case basis, rather than in AI technologies as a





whole. Studies have suggested that heatmaps may increase trust and contribute to the ethical use of AI [11]. This may be of particular importance when DL AI systems are being integrated into the clinical setting to be used in clinical decision support, where there is much discussion regarding the ethical implications of the use of AI in health care mooted in the professional



Fig 8. Participant agreement with binary AI diagnosis for all datasets.

https://doi.org/10.1371/journal.pdig.0000560.g008





literature [31,32]. There have been a number of methods of AI explainability proposed, with heatmaps indicating the area of focus of the AI in making its decision a popular choice [12,33].

As expert interpreters of radiographs the participants in this study disagreed with the heatmaps almost half the time (45.8%, n = 66). They indicated perfect agreement with the heatmap on 22.2% of occasions. Previous research has found that the visual feedback form of AI



Fig 10. Decision Switching: Participants' perception of their likelihood to change their mind from their initial diagnosis following *binary* feedback from the AI model.

https://doi.org/10.1371/journal.pdig.0000560.g010

Table 1	. Corre	lation	anal	ysis.
---------	---------	--------	------	-------

	INDEPENDENT	DEPENDENT	Correlation (Kendall's tau and Spearman's rank respectively) Effect size: small .01, medium .30, and large .5 (Cohen, 1988)
Correlation between agreement with <i>image heatmap</i> and <i>trust</i> n = 645 decision points	Agreement with HM (perfect (1), partial (2), disagreement (3)) n = 645 (decision points)	Trust perception (0–5, 0 representing <i>no trust</i> and 5 <i>absolute trust</i>) $n = 645$ (decision points)	515;—.584, significant large negative correlation at 0.01 level (p = < .01)
Correlation between agreement with <i>binary AI feedback</i> (HM and binary) and <i>trust</i> n = 216 decision points	Agreement with binary feedback (yes (1), partly (2), no (3))	Trust perception (0–5, 0 representing <i>no trust</i> and 5 <i>absolute trust</i>) following <i>all</i> AI feedback (HM and binary diagnosis)	309;—.369, significant medium negative correlation at .01 level (p = < .01)
Correlation between propensity to decision switch and trust $n = 216$ decision points	Decision switching perception (yes (1), maybe (2), no (3))	Trust perception $(0-5, 0$ representing <i>no trust</i> and 5 <i>absolute trust</i>) following <i>all</i> AI feedback (HM and binary diagnosis)	007,—.009, <i>not</i> significant at .05 level (p = .902, .899)
Correlation between image quality and agreement with binary AI feedback. n = 216 decision points	Image quality (0 –undiagnostic, 5 – excellent diagnostic quality)	Agreement with binary feedback (yes (1), partly (2), no (3))	238;—.268, significant small negative correlation at .01 level (p = < .01)
Correlation between <i>image quality</i> and <i>trust</i> n = 216 decision points	Image quality (0 –undiagnostic, 5 – excellent diagnostic quality)	Trust perception (0–5, 0 representing <i>no trust</i> and 5 <i>absolute trust</i>) following <i>all</i> AI feedback (HM and binary diagnosis	.219; .256, significant small positive correlation at .01 level (p = < .01)
Correlation between <i>image quality</i> and propensity to <i>decision switch</i> <i>n</i> = 216 decision points	Image quality (0 –undiagnostic, 5 – excellent diagnostic quality)	Decision switching perception (yes (1), maybe (2), no (3))	.086, .094, <i>not</i> significant at .05 level (p = .161, .167)

systems, using heatmaps, is coarse, at best [34]. Other studies have found that heatmap feedback may not be as useful as hoped to clinical end-users, including radiographers, who may find provision of the performance of the system a preferred metric to improve trust [35].

Whilst the user may prefer a different form of AI feedback, this study found that there was a large, significant ($\tau_b = -.515$, $r_s = -.584$, p < .01) negative correlation between trust and extent of agreement with the heatmap, i.e., when agreement decreased (from 1 representing perfect agreement and 3 representing total disagreement), trust perception also decreased. This is important to consider in user feedback/interface design.

5.3 Agreement of participants with binary AI diagnosis

The participants in this study reported agreement with the AI diagnosis on 86.7% of examinations (n = 26 out of 30 examinations). Previous studies have found that users may find heatmaps confusing, leading to excessive decision switching, and binary feedback useful, leading to increased accuracy [36]. The use of expert clinicians as participants in this study allowed for the responses to be determined as reliable, and the consensus opinion as 'ground truth'. Interestingly, in cases where there was disagreement with the AI, the participants noted sub-average image quality. The established diagnosis on the images included in this study are not available (Personal communication Rajpurkar, 2019) and therefore we cannot exclude the possibly that the participants may not agree with the established diagnosis. Further study into failure analysis should be conducted.

As expected, the extent of agreement of the user with the binary feedback from the AI was correlated with trust ($\tau_b = -.309$, $r_s = -.369$ (moderate), p = < .01) i.e. the greater the extent of agreement, the greater the perception of trust.

The binary diagnosis was presented to the participants following provision of all images and heatmaps in an examination, as would be the case in the clinical setting. Studies have found that the timing of the AI feedback may impact on the propensity of the user to follow advice from the system, resulting in confirmation and anchoring biases and potentially resulting in diagnostic error [36]. It has been suggested that the provision of any form of AI feedback should be optional, much in the same way as human-to-human feedback, however users should be aware of other biases with this method, such as confirmation bias [36]. Further research should be conducted into the timing of provision of differing forms of AI feedback on user trust.

5.4 Decision switching

Decision switching can be both positive, i.e., can cause the user to become more 'correct' or negative, causing the user to reach an incorrect conclusion. This has been highlighted as a potential issue when using technology to help make decisions, for example reliance on a spellcheck function, or in medicine, on clinical decision support systems [37,38]. Studies in medicine, including cardiology and radiography have found that the level of experience of the user may impact their likelihood to follow the advice of the decision support system over their own decision, with inexperienced users more likely to follow the automated advice over their own [37,38]. The participants in this study were all 'experts' in reporting radiography, having both clinical and academic training, and were all currently practicing in the field. No statistically significant correlation was found between rates of decision switching and trust in this study. This may be due to the expert users having less propensity to change their mind from their initial decision, which is supported by the aforementioned studies in the area. The overall perception of the likelihood of decision switching as a result of the feedback from the AI was low, with only two instances across all datasets where participants indicated that they would change their mind in response to the AI feedback (1.1%). In most cases (n = 179, 83.5%) participants stated that they would not wish to change their initial diagnosis.

5.5 Image quality

Users' perception of the diagnostic quality of each image was obtained initially as a means of quality assurance, although this also permitted analysis of the impact of image quality on likelihood to change their mind in response to the AI feedback and trust in the AI. There was no correlation found between image quality perception and decision switching, although there was a statistically significant mild positive correlation between image quality and trust in AI, indicating that as image quality increased, the trust in the AI also increased. This may have importance in the clinical adoption of AI tools, where the user may be negatively impacted by poor image quality from a diagnostic perspective, as reported in other studies [39] and also have a reduced trust in any AI tool adopted to help in the decision-making process.

Finally, a statistically significant positive correlation was found between image quality and trust (n = 216, τ_b = .219 (small), r_s = .256, p = < .01). This may indicate that when the user cannot clearly discern the area of interest highlighted by the AI that they are less likely to trust it. This may be an issue in situations where the AI proves to be more accurate in diagnosis that the human user due to degraded image quality. Future studies should compare the performance of human interpreters and AI for a variety of image resolutions. This may be particularly important when poor quality images are produced due to patient condition or suboptimal technology used to acquire the images.

There were no statistically significant correlations between trust and propensity to change their mind from the initial decision, or between the quality of images and propensity to change their mind.

5.6 Implications for practice and recommendations

Participants in this study agreed more often with the binary AI feedback than the GradCAM heatmaps provided. This is supported by the correlation analyses suggesting that the users' trust was impacted by their level of agreement with the AI feedback. There was a stronger correlation found in trust levels with the heatmap feedback than the binary diagnosis. Developers should be aware of this when creating forms of explainable AI and further study should be undertaken to investigate the impact of different forms of AI feedback on users' trust, with particular cognisance of the potential issues regarding the use of current forms of explainable AI on the human end-user [36,40].

Decision switching was not found to be linked to trust, perhaps indicating that users in this study remain anchored to their initial decision, regardless of their levels of trust in the system, which is supported by very low levels of decision switching. Education in the functionality of AI systems and their use may allow expert users to leverage the benefits of the system to increase diagnostic accuracy. Again, developers should consider the optimal means of communicating the feedback from the system to the end users. This may mean provision of a range of formats which the clinician can use to calibrate their trust in the system for each individual task, therefore allowing flexibility in its use on a case-by-case basis.

This study should be repeated with a larger sample of clinical staff of varying expertise levels and students at different points in their educational careers, to assess the effect of AI feedback on decision switching and trust. This would allow a more comprehensive assessment of the training and awareness issues, based on clinical experience and provide evidence for development of useful interface design and explainability methods to develop appropriate trust amongst end users.

5.7 Limitations and suggestion for further research

Whilst this study recruited an adequate number of participants, as per current recommendation in the literature [27,28] further research should be conducted with a larger number of participants with a varied background in the production of formal diagnosis from radiographic image interpretation, such as radiologists and other healthcare practitioners who provide formal written reports as part of their role.

The participants in this study were recruited from the UK only, which was intentional in an attempt to ensure that their role and educational background was similar. However, this may mean that the findings from this study may be generalisable to countries with high income economies and similar education and practice requirements only. Studies by Amugongo et al., 2032 [41] and Ewuoso et al., 2023, [42] note that the African (and, by extension potentially other countries with low to middle income economies) perspective on the concept of 'trust' may differ. Both studies suggest that, in this context, trust may be based on collective, experiential-based factors. This is beyond the scope of this study but should be considered in the context of potential disparities in cultural norms related to trust in AI and other emerging technologies.

Additionally, consideration should be given to long-term follow up of this study in this, and similar populations after a period of time using AI technologies in clinical practice to investigate any change in perception over time.

6 Conclusion

Clinicians have proposed that a lack of trust is a significant barrier to the successful implementation of AI systems in the clinical setting [10] (Fazal et al., 2018). This study aimed to clarify some of the factors impacting expert users' trust in AI systems for diagnosis of abnormality from radiographic images. The extent of user agreement with the AI heatmap and binary feedback has a positive impact on trust. Participants in this study did not agree with the localisation of GradCAM feedback, although on most occasions agreed with the binary feedback from the same AI system on the same images. This may indicate that users prefer binary feedback over this type of visual feedback and that disagreement with any form of AI feedback has a negative impact on trust perception. This will be important when designing the optimal user interfaces and forms of feedback for clinical use, where appropriate trust will ensure neither over- nor under-reliance.

Supporting information

S1 File. Study transcript-example image set from examination presented to participants. (DOCX)

S2 File. Per-participant diagnosis/examination and agreement with AI diagnosis. (DOCX)

Author Contributions

Conceptualization: Clare Rainey, Raymond Bond, Jonathan McConnell, Ciara Hughes, Sonyia McFadden.

Data curation: Clare Rainey, Raymond Bond, Jonathan McConnell, Ciara Hughes, Sonyia McFadden.

Formal analysis: Clare Rainey.

- **Funding acquisition:** Clare Rainey, Raymond Bond, Jonathan McConnell, Ciara Hughes, Sonyia McFadden.
- **Investigation:** Clare Rainey, Raymond Bond, Jonathan McConnell, Ciara Hughes, Sonyia McFadden.
- Methodology: Clare Rainey, Raymond Bond, Jonathan McConnell, Ciara Hughes, Sonyia McFadden.
- Project administration: Clare Rainey, Sonyia McFadden.

Resources: Clare Rainey.

Software: Devinder Kumar.

Supervision: Raymond Bond, Jonathan McConnell, Ciara Hughes, Sonyia McFadden.

Validation: Clare Rainey.

Writing - original draft: Clare Rainey, Devinder Kumar.

Writing – review & editing: Clare Rainey, Raymond Bond, Jonathan McConnell, Ciara Hughes, Sonyia McFadden.

References

- Lodwick G, Keats TE, Dorst JP. The coding of roentgen images for computer analysis as applied to lung cancer. Radiology, 1963; 81 (2) [online] Available from: https://pubs.rsna.org/doi/10.1148/81.2.185 [Accessed 15th June 2019] PMID: 14053755
- NHS. The NHS Long Term Plan. 2019a; Available from: https://www.longtermplan.nhs.uk/publication/ nhs-long-term-plan/ [accessed 30/03/23]

- SECTRA. The radiologists' handbook for future excellence: 2020. Available from: https://medical. sectra.com/resources/the-radiologists-handbook-for-future-excellence-2020/ [accessed: 3rd June 2020]
- Erickson BJ, Korfiatis P, Akkus Z, Kline TL Machine Learning for Medical Imaging. Radiographics, 2017; 37(2), 505–515 https://doi.org/10.1148/rg.2017160130 PMID: 28212054
- 5. NHS. The Topol Review: Preparing the healthcare workforce to deliver the digital future. 2019b; Available from: https://www.hee.nhs.uk/our-work/topol-review [accessed 24th June 2020]
- Greenspan H, San José Estépar R, Niessen WJ, Siegel E, Nielsen M. Position paper on COVID-19 imaging and AI: From the clinical needs and technological challenges to initial AI solutions at the lab and national level towards a new era for AI in healthcare. Medical Image Analysis. 2020; 66. https://doi.org/ 10.1016/j.media.2020.101800 PMID: 32890777
- 7. Badgeley MA, Zech JR, Oakden-Rayner L, Glicksberg BS, Liu M, Gale W, et al. Deep Learning Predicts Hip Fracture using Confounding Patient and Healthcare Variables NPJ Digital Medicine, 2018; 31.
- Qin C, Yao D, Shi Y, Song Z Computer-aided detection in chest radiography based on artificial intelligence: a survey. BioMed. 2018; 17(113). https://doi.org/10.1186/s12938-018-0544-y
- Guan Y, Wang X, Li H, Zhang Z, Chen X, Siddiqui O. et al., Detecting Asymmetric Patterns and Localizing Cancers on Mammograms. Patterns 2020; 1(7). <u>https://doi.org/10.1016/j.patter.2020.100106</u> PMID: 33073255
- Fazal MI, Patel ME, Tye J, Gupta Y The past, present and future role of artificial intelligence in imaging. European Journal of Radiology, 2018; 105, 246–250. https://doi.org/10.1016/j.ejrad.2018.06.020 PMID: 30017288
- Geis JR, Brady A, Wu CC, Spencer J, Ranshaert E, Jaremko JL, et al. Ethics of artificial intelligence in radiology: a summary of the joint European and North American multi-society statement. Journal of the American College of Radiology, 2019; 293(2), pp 1–6.
- Kumar D, Wong A, Taylor GW Explaining the unexplained: a Class-Enhanced Attentive Response (CLEAR) approach to understanding deep neural networks. 2018; Available from: <u>https://ieeexplore.ieee.org/Xplore/home.jsp</u> [Accessed: 10th August 2019]
- Cheng CT, Chou C, Liao CH, Chung IF, Ho T, Lee T, et al. Application of a deep learning algorithm for detection and visualization of hip fractures on plain pelvic radiographs. European Radiology, 2019; 29 (10), 5469–5477. https://doi.org/10.1007/s00330-019-06167-y PMID: 30937588
- Blüthgen C, Becker AS, Vittoria de Martini I, Meier A, Martini K, Frauenfelder T Detection and localization of distal radius fractures: Deep learning system versus radiologists. European Journal of Radiology, 2020; 126, 108925. https://doi.org/10.1016/j.ejrad.2020.108925 PMID: 32193036
- NHS. Diagnostic Imaging Dataset Statistical Release. 2020a; Available from: https://www.england.nhs. uk/statistics/wp-content/uploads/sites/2/2020/07/Provisional-Monthly-Diagnostic-Imaging-Dataset-Statistics-2020-07-23.pdf [accessed 2nd July 2020]
- 16. Olczak J, Fahlberg N, Maki A, Ali SR, Jilert A, Stark A, et al. Artificial intelligence for analyzing orthopedic trauma radiographs: Deep learning algorithms—are they on par with humans for diagnosing fractures? Acta Orthopaedica, 2017; (6), 581.
- Rainey C, McConnell J, Hughes C, Bond R, McFadden S Artificial intelligence for diagnosis of fractures on plain radiographs: A scoping review of current literature. Intelligence-Based Medicine 2021a; (5) https://doi.org/10.1016/j.ibmed.2021.100033.
- National Health Service. Diagnostics: Recovery and Renewal. 2020b; Available from: https://www.england.nhs.uk/wp-content/uploads/2020/10/BM2025Pu-item-5-diagnostics-recovery-and-renewal.pdf [Accessed 28th March 2022]
- The College of Radiographers. CoR Diagnostic Radiography Workforce Census 2020. Available from: https://www.sor.org/learning-advice/professional-body-guidance-and-publications/documents-and-publications/reports-and-surveys/or-diagnostic-radiography-workforce-uk-census-2020 [accessed 29/ 03/23]
- 20. The Society of Radiographers. Radiographer reporting grows as NHS struggles with radiology staff shortages. 2020; Available from: https://www.sor.org/news/government-nhs/radiographer-reporting-grows-as-nhs-struggles-with#:~:text=Radiographer%20reporting%20grows%20as%20NHS% 20struggles%20with%20radiology,a%20survey%20by%20the%20Royal%20College%20of% 20Radiologists. [accessed 29/03/23]
- The Royal College of Radiologists. Clinical Radiology UK Workforce Census 2020 report. 2021; Available from: https://www.rcr.ac.uk/publication/clinical-radiology-uk-workforce-census-2020-report [accessed 29/03/23]
- 22. NHS England. Draft: Diagnostic imaging reporting turnaround times. NHS. 2023; Publication reference: PR1753

- 23. Rajpurkar P, Irvin J, Bagul A, Ding D, Duan T, Mehta H, et al. MURA: large dataset for abnormality detection in musculoskeletal radiographs. 2018; Available from: https://arxiv.org/abs/1712.06957 [accessed: 15th May 2020]
- 24. Health Research Authority (HRA). Public Involvement. N.D.; Available from: https://www.hra.nhs.uk/ planning-and-improving-research/best-practice/public-involvement/ [accessed 29/03/23]
- Streiner DL, Norman GR, Cairney J Health Measurement Scales: a practical guide to their development and use. 2015; 5th Edition. Oxford University Press, England.
- 26. Fleiss JL, Levin B, Paik MC Statistical methods for rates and proportions (3rd ed.). 2023; Hoboken, NJ: Wiley.
- Obuchowski NA How many observers are needed in clinical studies of medical imaging? AJR Am J Roentgenol. 2004; 182(4). 867–9. http://doi.org/10.2214/ajr.182.4.1820867
- Allyn and Bacon.Participants, subjects and sampling2008; Available from: http://people.uncw.edu/ caropresoe/EDN523/523_Spring_08_Spring_09/McM_Ch5-Rv.ppt. Accessed 28th June 2021.
- 29. Pallant J SPSS Survival Manual (3rd Edition) 2007. Open University Press/McGraw-Hill. Berkshire.
- 30. Altman DG Practical statistics for medical research. 1999; New York: Chapman & Hall/CRC Press.
- Murphy K, Di Ruggiero E, Upshur R Willison DJ, Malhotra N, Ce Cai J, Malhotra N, Lui V, Gibson J Artificial intelligence for good health: a scoping review of the ethics literature. BMC Med Ethics 2021; 22, 14. https://doi.org/10.1186/s12910-021-00577-8
- Richardson JP, Smith C., Curtis S, Watson S., Zhu X., Barry B, et al. Patient apprehensions about the use of artificial intelligence in healthcare. npj Digit. Med. 2021; 4, 140. https://doi.org/10.1038/s41746-021-00509-1
- Viton F, Elbattah M, Guérin J-L, Dequen G. Heatmaps for visual explainability of CNN-based predictions for multivariate time series with application to healthcare. IEEE Int. Conf. Healthc 2020; 1–8. <u>https://doi.org/10.1109/ICHI48887.2020.9374393</u>
- Saporta A, Gui X, Agrawal A, Pareek A, Truong SQH, Nguyen CDT et al. Benchmarking saliency methods for chest X-ray interpretation. Nature Machine Intelligence 2022; 4, 867–878. <u>https://doi.org/10.1038/s42256-022-00536-x</u>
- Rainey C, O'Regan T, Matthew J, Skelton E, Woznitza N, Chu K-Y, et al. UK reporting radiographers' perceptions of AI in radiographic image interpretation: Current perspectives and future developments. Radiography. 2022; 28 https://doi.org/10.1016/j.radi.2022.06.006 PMID: 35780627
- Gaube S, Suresh H, Raue M, Merritt A, Berkowitz SJ, Lermer E, et al. Do as AI say: susceptibility in deployment of clinical decision-aids. NPJ Digital Medicine. 2021; 19; 4(1):31. <u>https://doi.org/10.1038/s41746-021-00385-9 PMID</u>: 33608629
- Goddard K, Roudsari A, Wyatt JC Automation bias: empirical results assessing influencing factors. International journal of medical informatics. 2014; 83(5), pp.368–375. https://doi.org/10.1016/j.ijmedinf. 2014.01.001 PMID: 24581700
- Bond RR., Novotny T, Andrsova I, Koc L, Sisakova M, Finlay D, Automation bias in medicine: The influence of automated diagnoses on interpreter accuracy and uncertainty when reading electrocardiograms. Journal of electrocardiology. 2018 51(6), pp.S6–S11. https://doi.org/10.1016/j.jelectrocard. 2018.08.007
- Mabotuwana T, Bhandarkar VS, Hall CS, Gunn ML. Detecting Technical Image Quality in Radiology Reports. AMIA Annu Symp Proc. 2018 Dec 5; 2018:780–788. PMID: <u>30815120</u>; PMCID: PMC6371374.
- Jabbour S, Fouhey D, Shepard S, Valley TS, Kazerooni EA, Banovic N, Wiens J, Sjoding MW. Measuring the Impact of AI in the Diagnosis of Hospitalized Patients: A Randomized Clinical Vignette Survey Study. JAMA. 2023; 330(23):2275–2284. https://doi.org/10.1001/jama.2023.22295 PMID: 38112814
- Amugongo LM, Bidwell NJ, Corrigan CC. Invigorating Ubuntu Ethics in Al for healthcare: Enabling equitable care. In Proceedings of the 2023 ACM Conference on Fairness, Accountability, and Transparency (FAccT '23). Association for Computing Machinery. 2023; New York, NY, USA, 583–592.
- 42. Ewuoso C Black box problem and African views of trust. Humanit Soc Sci Commun 2023; 10, 705. https://doi.org/10.1057/s41599-023-02213-9